Author’s response to reviews

Title: Capacity of Middle Management in Health Care Organizations for Working with People--A Case of Slovenian Hospitals

Authors:

    BRIGITA SKELA SAVIC (bskelasavic@vszn-je.si)
    ANDREJ ROBIDA (andrej.robida@telemach.net)

Version: 2 Date: 31 August 2012

Author’s response to reviews:

Dear Editor,

I made revision as you suggested:

1. Please include a title page as page 1. The title page should include the title, authors, author affiliations and corresponding author details.

2. Please include the Abstract on page 2. The Abstract should be composed of the following four sections and should not be more than 350 words: Background, the context and purpose of the study; Methods, how the study was performed and statistical tests used; Results, the main findings; Conclusions, brief summary and potential implications.

3. Please include a competing interests section at then end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

4. Please include an Authors’ Contributions section at the end of the manuscript, before the reference list.

kind regards, Brigita Skela Savi#